COLÁISTE NÁISIÚNTA EALAÍNE IS DEARTHA NATIONAL COLLEGE OF ART AND DESIGN FACULTY OF EDUCATION

B.A.DEGREE IN ART AND DESIGN EDUCATION

DISSERTATION ABSTRACT

STUDENT : ANN DEMPSEY

TITLE : ' USING THE IRISH MUSEUM OF MODERN ART A SUPPORT STUDY WITHIN THE JUNIOR CERTIFICATE COURSE '.

DESCRIPTION:

This dissertation is concerned with the integration of support studies into practical Junior Certificate Projects; I will refer specifically to utilizing galleries and museums. Throughout this dissertation I emphaise the importance of taking students to visit a museum / gallery. In this case I aim to show that the Irish Museum Of Modern Art in particular is an establishment worth exploring and utilizing as a source for support studies.

I will also discuss how it should be used to its full potential.



M0054410NC







COLÁISTE NÁISIÚNTA EALAÍNE IS DEARTHA NATIONAL COLLEGE OF ART AND DESIGN FACULTY OF EDUCATION

' USING THE IRISH MUSEUM OF MODERN ART AS A SUPPORT STUDY WITHIN THE JUNIOR CERTIFICATE COURSE'

A DISSERATION SUBMITTED TO THE FACULTY OF EDUCATION

IN CANDIDACY FOR THE

B.A. DEGREE IN ART AND DESIGN EDUCATION

BY ANN DEMPSEY

1998

COLLAISTE NAKULINTA EALANGIG DEARTMA NATIONAL DOLLEGE OF ARTAND DESIGN FACULTY OF EDUCATION

USING THE IRISH MUSEUM OF MODERN ART AS A SUPPORT STODY WITHIN THE JUNIOR CERTIFICATE COURSE!

A DISSERATION SUBMITTED

IN CANELDACY FOR THE

MORYADU DE PERSENTAND DESEDRIES LICATION

BY ANN DEMPSE

CONTENTS

LIST OF ILLUSTRATIONS	IV
ACKNOWLEDGEMENTS	v
INTRODUCTION	VI
CHAPTER	

1. SUPPORT STUDIES IN THE JUNIOR CERTIFICATE 1

Art, Craft and Design within the Junior Certificate Syllabus A closer look at the Junior Certificate Course Support Studies within the Junior Certificate Course Support Studies : Whats it all about ? Primary and Secondary Sources

2. WHAT A MUSEUM HAS TO OFFER

8

What is a museum / gallery ? Why use the museum / gallery as a source ? The importance of taking students to a museum / gallery What a museum / gallery has to offer The importance of a worksheet The importance of a educational officer

3. INTEGRATING THE MUSEUM INTO PRACTICAL A EXERCISE 18

Background information on the school Outline of the scheme Background information on IMMA Andy Warhol exhibition in IMMA

SUMPLY OD

LIST OF CLUDETSATIONS ACTOMILEDOENENTS

STADENTS NOR SATURATION OF THE START REPORTS

An Greater (1999) with the antiple Dentities (1989) A floed (1911) the Junior Dentities (Color Sectors) Support (1994) with a the Junior Confecto Courses (1966) and patients Sources (1997)

MAATA MOSELIA BASTO DEED

vinar is an over 7 gallery ? "Invuse Procuses ray gallery as 3 source ? Procusponance of aking students to a molecum / gallery m vin a ray a ray gale y has to ong Procoder of the viol office. Procoder of the viol office.

A EXCEPTION THE PUBLIC PRACTICAL PRACTICAL AND PRACTICAL A

4. PLANNING A VISIT TO A MUSEUM / GALLERY 34

Planning a visit to IMMA Preliminary work for a museum / gallery visit Preliminary work within the classroom Actual visit to IMMA Follow up work back in the classroom

5. CONCLUSION	54
APPENDICES	59
SELECTED BIBLIOGRAPHY	63

PLANMING A VISIT TO/A MUSEUM/ GALLERY 3

Prolimitary work for emuseum / gallary visit Prolimitary work for emuseum / gallary visit Prolimitary continent on plassroom Arthol visit to 1971A

8 COMCLUSION APPENDICES

SELFOTED FIBLIOGRAPHY STATES

LIST OF ILLUSTRATIONS

ILLUSTRATION	PAGE
1. Line drawing of the face by Picasso.	24
2. Line drawing of the face by Matisse.	25
3. Slides used as a form of support studies.	26
4. Self portrait by Andy Warhol.	27
5. Silk screen print of Marilyn Monroe by Warhol.	28
6. A portrait by Peter Howson.	29
7. Warhol has used various colour combinations	
to convey the image of the face.	30
8.Students at work in the museum.	38
9. Students at work in the museum.	39
10.Student making a tonal drawing of their face.	44
11. Student painting the large A3 size portrait of	
their face.	45
12.Student getting ready to trace.	46
13.Student tracing out stencil.	47
14.Students cutting out stencils.	48
15 Students at work on their screenprint -	
dragging the ink.	49
16.Students at work on their screenprint - lifting	
the screen.	50
17.Examples of students first stencil prints.	51
18.A finished piece of work.	52

LIST OF ILLUSTRATIONS

ILL'ISTRATION
2.1 me drawing of the face by Matisee.
acodemidmoo huoiga suontry bead estitutin Mid

ACKNOWLEDGEMENTS

I wish to thank the following individuals for their valued assistance and support which has culminated in the writing of this disseration.

Maria Farrell, my tutor, for guidance in research, organization of material and for the discussions which enabled me to clarify my ideas.

I also wish to thank the following people ;Museum professionals in IMMA for their particular and valued assistance, to the staff and students of Mount Temple Comprehensive School who were cooperative and enthusiastic in their participation in the research scheme.

Finally, I am indebted to my family and friends, and in particular my parents, who have been so supportive and encouraging over the past number of years - to them I am very grateful.

ACKNOWLEDGEMENTS

Invisit of react the following individuals for their case match support which the following individuals for the dissertation.
Invisit for a submatchin the writing of the dissertation.
Invisit for the table of the writing of the dissertation.
Invisit for the table of the writing of the dissertation.
Invisit for the table of the writing of the dissertation.
Invisit for the table of the writing of the dissertation.
Invisit for the table of the table of the writing of the dissertation.
Invisit for the table of the table of the dissertance of the dissertance.
Invisit for the table of the table of the dissertance of the dissertance.
Invisit for the table of the table of the dissertance.
Invisit for the dissertance of the dissertance of table of the dissertance of the dissertance.
Invisit for the dissertance of the dissertance of table of the dissertance of t

ensive some indepted for my sonally and mondal and to particular pay or rendo twing have been ad supportive and encourring over the past or hor of yours or them that very grateful

INTRODUCTION

This dissertation is concerned with the integration of support studies into practical Junior Certificate Projects ; I will refer specifically to utilizing galleries and museums. I aim to show and prove that the Irish Museum Of Modern Art in particular, is an establishment worth exploring and utilizing as a source for support studies.

Chapter one deals with what support studies is and how it fits into the Junior Certificate Syllabus framework. By looking at this framework and the Junior Certificate Syllabus we will achieve a greater understanding in the area of support studies.

Chapter two looks at what a museum is and the importance of bringing students to a museum. I will also discuss what a museum / gallery has to offer in terms of worksheets and educational officers.

Chapter three looks at the Irish Museum Of Modern Art as an educational resource and entails an account of the scheme itself, in which I used the Andy Warhol exhibition as a support study.

Chapter four examines the stages involved in organising a visit to IMMA from preliminary preparation through the visit and follow up work in the classroom.

Chapter five deals with my own personal evaluation and conclusion, which are summed up in this short chapter.

MOLIOPEDORTM

The distantiation is provented with the Integration of support studes, and practices functing differences in trial treffic specificacy to motion generies for minimum dient to show and prove that the train dues in the original and managed to show and prove that the train dues in subsorper as a courter re-structure establishment worth and original comment memories who what show a structures is and how of his into the due to be reacted to structure for the train memory and provides in and how of his into the line to be reacted at moving the provides in and how of his into the line to be the courter structure of the top of a structure for the structure structure is and the moving and due to be structures at moving is and the moving and during the structure of the structure of the structure of subsorptions are a structure of the moving during the structure of the structure of the moving structure of the structure of subsorption and structure of the moving structure of the structure of subsorption and structure is and the moving structure of the structure of subsorption and structure of the moving structure of the structure of subsorption and structure of the moving structure of the structure of subsorption and structure of the moving structure of the structure of subsorption and structure of the moving structure.

evoluer in terma ou voi keneetă ano educational pilicers. O iapren driaa nou la tra Insin Masaum Or Modera Africas, en colarene erecine asin receltik sero-coourt of the dolarie deal, un vrinni asid file Ando Marina existik sero-coourt of the dolarie deal, un creater four exemitor dia scores disorrazea algon dia indi creater four exemitor dia scores disorrazea algon dia notificative from pretionary preparation financial de data endo contra pilato. Por pretionary preparation financial de data endo contra pilato.

Charren five deals with my providers and evaluation and ponchaion.

CHAPTER ONE

Art, Craft and Design within the Junior Certificate Syllabus.

The Junior Certificate Art, Craft and Design Course was set up in 1987 by the National Council for Curriculum and Assessment (NCCA). The Junior Certificate replaced the Intermediate Certificate and was first examined in 1992. Like the Intermediate Certificate it is a three year course although the layout of both differ greatly. The Junior Certificate is a much more stimulating and exciting approach to the Art, Craft and Design Curriculum which makes it more interesting for both the teacher and student.

It sets out to give a broad understanding of Art, Craft and Design at different comprehension levels. The programme aims to reinforce and develop the educational expierence in terms of knowledge, understanding and skill. It also aims at developing the young persons personal and social confidence and contributes to the moral and spiritual development of the young person as they develop an open mindness and respect for the values and beliefs of others. The Junior Certificate prepares the student to further develop their programme of study and can prepare them for employment or life outside full time education. The curriculum should provide a large range of educational expierences within a helpful and formative environment;

1

BHO SETAND

in, Craft and Deuten villanthe Junior Certificate

Condition of the sets and to give a coold or de clanding of Art (Cod Stand and the althout complete provide the sets of the program althout manage of the sets of the seturation of the set provides and the set of conditions to contrain and and althout sets and antisipping the young conditions contrained instantic sets and contributes to the morel and provide a more the sets of the sets of contributes to the morel and set the sets of provide a set of the sets of the sets of the sets of the sets of provide a set of the sets of t It should build on the aesthetic and creative, the ethical, the linguistic, the matematical, the physical, the scientific and technological, the social, environment and political and the spiritual domains (1).

A closer look at the Junior Certificate Course.

The Syllabus is designed to ensure a good balance between expierencing, making and understanding. Art, Craft and Design is seen through Drawing, 2D and 3D and they are treated individually using Support Studies to back up their learning expierences. The course is composed of a core syllabus and a number of optional studies. The student can select from a variety of thirtyone options , ranging from Animation to Weaving. The choice of options is so diverse that the teacher must have the skill to teach the majority of these options and the school itself must provide the resources and facilities needed.

The Intermediate Certificate was the recognised examination previous to the Junior Certificate and students were examined in three areas - imaginative composition, still life and design. The Intermediate Certificate course placed great emphasise on the end product and assessment involved an examination on a specific day. This was a very restricting and limiting system. However, the Junior Certificate level students work in an assessment based situation in which they are required to apply the skills that they have developed over a three year cycle to in social auto de ancessi eso ana presensi, de annai me à relationa de resentation de parezar lite scieduro add de trabalação de social de company no postical and the

A closed took at the Janior Centificate Courses.

and model mainers of orderations of an unit and theory is are mean theorem and an analysis and theorem or and immutually using orderation States in that so and the fermine incorrences. The sectors compose also are spaced and the model of quarterines. The sectors are exact to the mean of mityers concrete much from Ammetor to the clinic tests for the order of these options and the most news the clinic tests for the order of these options and the resolution are the clinic tests for the order of these options and the resolution are the clinic tests for the order of these options and the resolution are the clinic tests for the order of these options and the resolution are the clinic tests for the resolution of the order of the order of the test of the resolution of the order of the order of the resolution are the clinic tests for the resolution of the order of the order of the order of the clinic tests for the order of the order of the order of the order of the clinic tests for the order of the order of the order of the order of the clinic tests for the order of the order of the order of the order of the clinic tests for the order of th

various art, craft and design tasks. This project - based system puts more emphasise on the process which enables students to spend a specific length of time completing their projects. It examines ongoing work which can help show the importance of process and then the product. This allows for full exploration and investigation of a chosen topic and provides a practical time scale which enables the student achieve the best results.

Support Studies within the Junior Certificate Course.

The general aim of the Junior Certificate Syllabus is;

to contribute towards the development of all aspects of the individual, including aesthic, creative, critical, cultural, emotional; intellectual, moral, physical, political, social and spiritual development for personal and family life, for working life, for living in the community and for leisure. (2).

The Junior Certificate allows so many different approaches and directions to be undertaken. One can see a new way of working, developing and learning which allows the student involved to benefit from this type of education through Art, Craft and Design. The Junior Certificate emphasises the importance of aesthetic studies. Realising the importance of aesthetic which was introduced in the form of Support Studies.

Support Studies adds to the aesthic expierence of a student and the integration of it into the Junior Certificate art course is essential and

Sanooli Studies weath the unlike Collings

.

one cannot emphasise it enough. The work of Eilot Eisner and his model of artistic learning has obviously influenced the decision to include support studies in the Junior Certificate Art Syllabus. He does not accept the point that artistic learning is an automatic consequence of maturation. He fully believes that art education can be facilated through instruction and direction. He states that ;

artistic learning is not a single type of learning. It deals with the development of abilities to create art forms, the development of powers of aesthetic perception, and the ability to understand art as a cultural phenomenon. (3).

It is important to look closely at Eisner's three aspects of learning;

(1) The Productive.

(2) The Critical.

(3) The Cultural.

If one looks at the first aspect which is the Productive, it involves technical elements, that is the actual making skills, techniques and materials. The Critical aspect deals with the making of judgements and being able to discuss your work where by the Cultural is seen as the historical dimension. One must try and get a balance between all three aspects of learning and the Junior Certificate certainly deals with all three. and a constant of a cough if he work of Fibir Ferrer and he control of musics canning has acreately influenced the Harlson to lock de scionon process of a locate he set of the of the split bus the fores of the could are could be accessed at a statement of the set of the could be statement of a set of the set of the set of the set of the could be statement be accessed at a statement of the set of the could be statement be accessed at a statement of the set of the could be statement be accessed at a statement of the set of the could be statement be accessed at a statement of the set of the set of the could be statement be accessed at a statement of the set of the set of the could be statement be accessed by the set of the set of

A state infinite a cost explicit was a submitted as the second explicit of the

is sentilities for a lease of the result of the sentilities of the

Support Studies : Whats it all about ?

The Junior Certificate Syllabus states that support studies focuses on ;

critical appraisal, history and appreciation of Art, Craft and Design, and related studies should be organised to form an integral part of the learning expierence in order that the student percieves new meanings and new discoveries . (4).

Support Studies helps the teacher to illustrate their point about any given topic and makes the students more critical about their own work. Students will have a better knowledge of their subject matter if the teacher reinforces the idea of support studies within the classroom even if it is only a support studies notebook. There is not enough emphasise on support studies. If the teacher presents to the students the work of artists either past or present which relate to the topic they are studying then support studies will give them a greater insight and understanding about this specific subject. The N.C.C.A. has stated that support studies can be related to ;

ideas, movements, working methods, influences, meanings, impacts, problems encountered and problems solved etc...(5) and these should be taught as an integral part of each learning expierence (6).

Support Studies are a vital part of the Art, Craft and Design Course. The process of the project is not only of educatioal value but it is important to see the final product as not just evidence of sucess or failure but as an

Support Singles - White all another

and a set of the second of the system of the support shall be a set of the second shall be a set of the second s

לי היי היי היי היי היי הייני הייניה איניה איניין הייניין הייני אוקר הייני הייניין הייניין הייני הייניין הייני הייניין הייני הייני הייניין הייני הייניה הייניה הייניה הייניה הייניה היינים הייניין הייניין הייניין הייניין הייניין הייניין הייניין הייניין הייני הייניה היינייה הייניה הייניה הייניה הייניה הייניה הייניה הייניה הייניה הייניה היינייה הייניה היינייה הייניה הייניה הייניה היינייה היינייה היינייה הייניה הייניה הייניה היינייה היינייה הייניה הייניה הייניה הייניה היינייה היינייה היינייה היינייה היינייה היינייה הייניה היינייה היינייה הייניי הייני היינייה הייניה היינייה הייני הייני היינייה היינייה היינייה היינייה היינייה היינייה היינייניייה היינייניייויוינייייי היינייה היינייה הייניייה היינייה

anadro proversi encol place or and problems advised of 2000

effort and structure that reveals to us valuable information about the way the idea originated, developed and was realised.

Primary and Secondary Sources.

The Junior Certificate Examination demands that support studies and primary and secondary sources are fully developed. Primary sources refers to using original pieces of art work from galleries and museums or art work by pupils or teachers. Secondary sources however, is a reproduction of art work like photoes, slides, postcards, prints and so on. The Junior Certificate prmotes students to begin working from a primary source and leave the secondary source to back- up the work.

A trip to a museum / gallery is seen as a primary source and a good form of support study. These places of interest contain wonderful pieces of work for both teachers and pupils to use. The main reason for visiting a museum / gallery is for pupils to see for themselves original works of Art, Craft and Design. Visits must be made for a real purpose and should focus on only a few pieces and the choice should carefully support and extend current classroom work in the school. The expierence of art should be broadened and by seeing an artist's finished work through visiting galleries, museums, workshops etc. is one way of doing this. and ended the terror reveals to at visit and the endourse meters of the second the second states of the second

Locies and has been to solve and many solve the

Primary and Secondary Sources

A Hoke form of the providence there information interest to the standard dependence of a loss for hole considered there is an experiment of hole considered to the providence of a loss for hole considered to the providence of a loss for hole considered to the providence of a loss for hole considered to the providence of a loss for hole considered to the providence of a loss for hole considered to the providence of the providence of a loss for hole considered to the providence of a loss for hole considered to the providence of the providence of a loss for hole considered to the providence of the pr

FOOTNOTES

- (1) Department of Education, National Council for Curriculum and Assessment <u>"Junior Certificate Art, Craft and Design</u> <u>Syllabus</u>" (Dublin:The Stationary Office,1992) p.1.
- (2) Ibid.
- (3) Eiliot Eisner, <u>Educating Artistic Vision</u> (New York:MacMillan Pub.,1972) p.66.
- (4) Department of Education, National Council for Curriculum and Assessment <u>"Junior Certificate Art,Craft and Design</u> <u>guidelines for teachers</u>" (Dublin,The Stationary Office,1992) p.1.
- (5) Ibid.
- (6) Ibid.,p.16.

ER CONTROLES

 DEPARTMENT of Education: National Council for Olymenium. and Assessment <u>Judio: Conflicter Art Total and Design</u>. Sylichysele Dublin: Gal Stationary Office (1992) off.

And a

Provide Barren, <u>Buildaring Artistic Vision</u> (New World and Sachilitan Park 1972) p.65

n Department of Education, rocional Council for Curronlum and Assectment, <u>Junior Conificate Art Craft and Doafon</u> auto <u>Stationan for teachers</u>" (Doblin The Stationary Office, 1992)

a sum of

20 C 19 C 19

CHAPTER TWO

What is a museum / gallery ?

A museum says the Internatioal Council of Museums is a ;

permanent establishment administered in the public interest, with a view to conserve, study, exploit by various means and basically to exhibit, for the pleasure and education of the public, objects of cultural value. (1).

A museum / gallery is one of the best primary sources available to students who are studying art. Many were built in the nineteenth century and were designed to educate people about specific artefacts and exhibits. These establishments were regarded as " historically valuable instructors "(2), were visitors went to learn. In a museum / gallery one expects that there will also be possibilities for the visitor not only to learn facts but also to have expierences we call aesthic. As Hans L. Zetterberg stated ;

Museums were concieved as treasuries of beauty. They should do for beauty what the central bank did for the wealth of the nation. The bank kept the deposited wealth safe and made it available for investment. In a similar vein, it was argued that museums should keep the deposited beauty safe and make it available to artists as inspiration to create new works of art. (3).

8

CHAPTER 14:0

What is earlied and up failery?

numero serve not stormation!"Council of Museums in a company server of meet, Scoremetered in the matterest, when however exection respiration and the respiration of the costanty to particle state the personance and information of the child subscored of and when

A you surv gallary relevand the best homany subcus and build all a sitularity on the standard and the best homany subcus and the balance of the best homan with a second control with the balance of t

reactive Volumbence of right for substance chartering worker

NE MENC

Nowadays, museums and galleries play a vital part in the community and are generally more active in seeking " collaboration, consultation and involvement from within their communities "(4).

Why use the museum / gallery as a source ?

The primary purpose and function of a museum / gallery and its exhibits is to educate. Therefore, these establishments are educational resources and pupils should be taught how to use them and they should be encouraged to explore them.

It is very easy for a student to be educated throughout their school career without leaving their classroom which is very limiting and restricting for both student and teacher. If I have a personal regret, it is that my own education was lacking in these such expierences, for I know now just how much one can acquire from working in this environment. A visit to a museum, gallery or site can be one of the most exciting and memorable parts of a person's education. Therefore, I feel students should be taken out of the classroom situation and make more use of the educational resources available.

Novuse fils mussum (gallery as a source ?

S HERENG 2500 DEET

The importance of taking students to a museum / gallery.

In relation to the teacher a lot of work and research is associated with organising a school trip from permission notes, phonecalls, preliminary visits to worksheets to actually bringing students out on the school trip. Teachers have to put together new schemes, learn new skills and take on additional duties and responsibilities. A teacher could lessen these presures by following simple procedures by approaching their local museums / galleries or by actually learning how to use a museum / gallery to its full potential.

It is important to use the visit to accomplish a well defined purpose and feed it into the students work rather than developing an isolated expierence. The art teacher must try create a feeling among the students that the only way to accomplish the next stage of their project is to visit a museum / gallery. The visit will be much more purposeful than just recording an account of the day as students will know in advance what they are looking for .The visit must be very focused so the student can use the information gathered to extend current studio work in the school.

Both student and teacher need to adapt to working in a new environment that is not their classroom. Students like to go on school trips as they see it as a challenge because they are visiting an unfamiliar environment. The students are so used to hearing information from the
The importance of taking students to a nurseum v

gatterys

teacher they are not used to seeing the actual piece of work in front of them. Rod Taylor describes a trip as being ;

something that is valuable to them , something which they can recall with some pleasure and feel that they really have made some breakthrough in their perception of art works .(5). A visit to a museum allows students to witness new expierences, and they have to ;

speak, listen, read, write and learn new vocabulary in order to cope with the visit and make use of it afterwards. (6).

One can refer back to the aims of the Junior Certificate where it reinforces this particular point and it places an emphaise on students need to develop a practical vocabulary. This will give the students confidence and increase their knowledge and understanding. So the visit is of great benefit for students and it is up to the teacher to make use of this expierence when they return to the classroom situation. Students now have learned how to ;

develop skills of observation, recording, analysis of recorded material and the synthesis of the results of analysis. They alsoneed skills in discussion, speculation based on evidence, explanation of ideas and arguments, constructing theories, testing them and modifying them in the light of subsequent discovery. (7).

Therefore, one can safely say that museums / galleries can provide the most desirable working environment for this exercise were one can improve and develop these skills. Increasing numbers of teachers are recognising

specie fister, end with sparinger new vocabularities tor to end

Construction of the state of the control of a dimensional of the second of the state of the second of the secon

n art daaraale violong enviorment for this such courses an a sub light offer and daaraale been diving tomesting of outpoters of the scheme such and the such as the importance of taking their students out of the classroom and into other learning environments. Museum / gallery visits are always likely to occur as a result of the enthusiasm, confidence and expertise of individual teachers. However, not all of them have the confidence or feel they have the expertise to take on such a chore while others find the task to demanding. Nevertheless, I am going to discuss the importance of bringing a class to a museum / gallery with specfic aims and objectives in mind. This practial investigation will involve using The Irish Museum Of Modern Art as a primary source to work from within a scheme.

What a museum / gallery has to offer.

In order to learn from their trip their teacher must use the museum / gallery to its full potential. Just taking them to the museum is in itself insufficient as teachers need special skills to set up the visit, allow it to take place and follow up museum work back in the classroom. Most museums will do their best to help the teacher use a museum to the best effect. Museums offer material for the best effect. They offer material for the use of both teachers and pupils alike, practical guidance such as opening times, costs, facilities available, site information, guidebooks, contents lists of galleries. background to the museum and lots more.

A second design of the disercet subsects of the question as a finite question as a finite question of the q

who a measure called interaction and

The importance of worksheets.

Museums / galleries offer worksheets and activity sheets for use during a visit. Prior to any visit to a museum / gallery, the teacher must talk through with the pupils what they are going to see and build up their sense of curosity. Students should be introduced to the idea of using sketchbooks and worksheets. Time spent with just one or two pictures or objects after a brief walk around a gallery can often prove far more valuable than trying to look at everything. When one is taking students to visit museums and galleries to see exhibitions as part of their work, it is essential to use well designed worksheets in order to help them focus upon particular works of art and related tasks within a complex exhibition. Worksheets need to be designed by the teacher specifically for their pupils and specifically for that visit if they are to be any good. Worksheets should contain a number of tasks and questions directed to the key facts. They should not be treated as an end product but must be used to further an investigation. Worksheets have to stimulate students to observe closely and to ask questions about what they see and to give the opportunity to gather the relvant information necessary to answer these questions. One must avoid anything too simple and worksheets must be well written, well planned, clear, illustrated, enjoyable yet educational. The worksheet must get students thinking not just simply reading the labels and filling in blanks.

One must remember that students like to roam around a museum / gallery and it is in these circumstances a worksheet would be a good idea. A worksheet can be constructive and investigative work in which

alconarian to again component

students skills and understandings are developed. Pupils are allowed work at their own pace and some will be more interested in certain areas than others. A worksheet can provide more structure to a lesson than a general loof around, giving a greater opportunity to learn and enjoy the expierence. A good worksheet is vital for the sucess of a museum / gallery visit.

The quality of worksheets available in museums / galleries vary from place to place. Some worksheets are successfully designed while others are to difficult or to general and not well thought out while some galleries offer none at all. As regards IMMA they have no worksheet designed for the Andy Warhol exhibition so it was up to myself to design my own worksheet if I wanted my visit to be extremely successful. This took a lot of time, effort and research, being selective and critical with information and questions in order to design a good worksheet. The worksheet emphasised relevant information to be collected by the students. The questions asked encouraged the students to explore each area of the paintings or prints, the materials, methods and techniques used and their relationship to other artworks within the exhibition. The worksheet also concentrated on how the artist treated the image of the face and his ideas behind his work and so on. Overall the worksheet was very clear, simple and extremly focused. (Appendix 1)

14

The objective of worksheets and actively in a barrier (gain dealers) in a second of the second second and actively the particular to the second of the particular to the second of the particular to the second second and actively the second second

The importance of educational officers.

As regards a visit to a museum / gallery one must concider how the educational officer can assist you. Sometimes teachers do not seek the museums advice educationally before a visit which is a great pity. Realistically the teacher should inform the staff what is required from the visit for example what will they be doing back in the classroom, what is the link, what is the ability and background of students and so on. This information is of great importance to the musuem / gallery staff because they can be more specific and can focus on key areas of exhibition. The educational officers are usally more than willing to tailor your visit to your specfic needs and will happily prepare suitable introductions to collections / suitable images, providing you give them enough warning. They also need to know how much preparation has been done prior to the visit and has the class much expierence in looking at pieces of art work and have they been to a museum / gallery before. The art teacher needs to know about the outcomes / objectives which needs to be achieved through the visit purpose of visit.

Educational officers are "subject specialists" (8). Most big museums and galleries have an educational officer. The museum which I am working with has a very good educational officer, Ann Davrin whom works within The Irish Museum Of Modern Art and whom is only happy to provide advice and resources to any one seeking information. One must remember not to take for granted that the education officer is going to do all the work, provide all the information, materials and manage the groups of

ou importance of actucational officers

Alow and executive officer and easier and Cometances Levanne (co. not now and executive officer and easier and Cometances Levanne (co. not here is the relation officer and the original form - Constrained from the each the relation of the variation of the original form of the original form of the relation of the variations double barrier of the original form for each the relation of the variations of another or the original for the relation of the variations of the original form of the original form of the original officer of the original form of the original form of the original officer of the original of the original form of the relation of the original of the original of the original for each or the specific of the original of the original of the original officer or officer or the original of the original of the original officer or officer or original form or original of the original officer or officer or original or the original of the original officer or officer or original form or original or the original original

Ecourar et el attrate de la secone en tracique apor alters (1892-Most utp in estima anté puis las lasse en aducatignates) de elucatione des un solo e vent sela ven puis d'attratione concel. Attra 2000 pérez vor ecoles end resonades toran Ottédopernént en divinnation activit recipi to provide advisa end resonades toran cons sections information. Occumination de ender en toran toran torgenales for any one section divisionation activitation de ender en toran torgenales toran cons sections information. Occumination students. This is not the educational officers job but the teacher in charges responsibility. Before any visit takes place the teacher in charge must get to know the educational officer as well as the actual museum / gallery. One must talk with them, let them know what you want and be prepared to listen to what they have to say. These people are trained in this area and know a lot about how to get the best out of the resources they have to offer. The educational officer can play a large role in making the visit to a museum / gallery a successful trip. If one cannot make a trip to the museum / gallery they might be able to send out one of their educational officers out to the school and discuss what is to offer in the museum / gallery through use of slides, videos, replicas and so on. This will not be as effective as the real thing but it is an alternative.

Another and the residence of the contract of a contract of a contract of the contract of th

FOOTNOTES

(1) Kenneth Hudson, <u>A Social History of Museums:"What the</u> vistors thought"(London:Reakton Book Ltd.,1989) p.1.

(2)Ibid.,p.3.

(3)Hans L.Zetterberg,<u>Museums and Adult Education</u> (London:Evelyn,Adams and Mackey,1968) p.4.

(4)Ibid.,p.16.

- (5)Rod Taylor, <u>Educating for Art:Critical Response and</u> <u>Development</u>(London:School Curriculum Development Committe,1986) p.133.
- (6)Graeme Talboys, <u>Using Museums as Educational Resources</u> (London:Ashgate Publishing Ltd.,1996) p.19.
- (7)Department of Education , <u>Charting our education</u> <u>future"White Paper on Education"</u> (Dublin:The Stationary Office,1992) p.126.
- (8) Talboys, Using Museums as Educational Resources, p.23.

131 OM 100-

(1) Kennoch Hudson, A. Scelal History of Missourist Wrightha Vistora throught "Condencies adden Book, 16, 1989 pct.

(2)":std., p. 35

(5)Ports L.2 distribution <u>Resolutes</u> and <u>Addit</u> <u>Education</u> (London Evely discussion Receipt 1953) p.4.

dr. , biegs)

State Carrier Economics to <u>NatConcol</u> Response <u>Fand</u> <u>Development</u> LongousSchool <u>Santoutum</u> Development Correcte 10.81 p.455

Drass of Albridge <u>Haund Musoums as Horoalionint Resources</u> LondontAssignate Publishing Links (981) p.43.

(7.6 marketern of Schuczung <u>Churang pur edecation</u> <u>Anne White Paper on Education</u> (Outrin Workshing) Office (9.00 m for

File above, Dama Museums as Sourceinner Roconnos, 0.23

CHAPTER THREE

Background Information on the school.

The school in which I carried out my research for my dissertation was situated on the north side of Dublin. It is a large comprehensive school with approximately six hundred students. The school evolved as a result of three schools merging, bringing together students of different social and religious backgrounds. The school is very liberal in that the students do not wear a uniform however there are limitations and a dress code is followed.

There is a very strong school spirit and this is clear from both students and staff. There are few disciplinary problems in the school. This is because of good management as well as good teacher pupil relationship. The staff are extremely friendly and very willing and interested to help. There is a strong support for the art department within the school and this support comes from all staff. The school has two art teachers who are very different in their approaches to teaching. There are also two art rooms, both of which are well set up and well equiped. During my teaching practice in this school I divised schemes of work for all my classes which laid particular emphasise on the whole area of support studies. In order to analyse the value that support studies has on a project I decided to select one scheme of work and carry it out in my teaching practice. Therefore, I am going to discuss the importance of using IMMA as a support study by implementing an actual museum / gallery visit to the Andy Warhol exhibition presently be shown. One cannot emphase enough the importance of

CHAPTER THREE

Background information enable school

The set of the monitors of the set of the standard of the set of t

Alternative of each of the objection of the second of the

support studies as it strengthens the students learning ability and a museum or gallery visit can be an excellent way of proving this point.

Outline of scheme.

The following is a brief outline of the scheme involved which I have already said was under taken in a comprehensive school on Dublins northside. This second year group were mixed and of mixed ability. There was a total of twentyfour students and the scheme ran for the duration of ten weeks and incorporated a visit to IMMA.

AIM : To develop an awareness and understanding of screenprinting with a view to creating personalised wallpaper on the theme self-portrait.

Week One :

Objectives : To create an awareness of the proportions of the face through practical investigation.

Description : The students spent the first class learning about the proportions of the face, for example were are the eyes found in relation to the nose or whats the distance between both eyes. Students practiced this by making several line drawings of the face indicating the correct proportions of the face. Students were also introduced to artists who dealth with the subject of portraiture in their work.

supplinit sudjest solar Misnightaris the studiet is reaming ability and museum or gritely rish can be an excellent way of moving this point

Outline of a cheme

the following is a submatting of the source involved which t

Non el controlado de la controla de la controla elevador d'autoritatione inclusione The secol o resolucione wave maxol and al anno trabilitation inclusione to swardiour autorita autorita esterne registra avaitationo i

Steek One

Dudulivon Monalia en avaranses d'ille volonne er les faces enough o actori mesitanta Despisoni : The statemic spectation vet muse resmund appul de

which the subject of contralutes in the

Week Two :

Objectives : To reinforce students knowledge of the proportions of the face while creating their own self portrait.

Description : Students now have a knowledge of the proportions of the face so they began to put this into practice and had to draw their own self portrait with the use of a mirror. Students referred back to last weeks work as a guide if they were not sure about were a certain feature was positioned. Students at this point began to assemble a support studies copy. The copy includes drawings of the proportions of the face , documented what they had learnt, stuck in photos of themselves and so on.

Week Three :

Objectives : To develop an understanding of the effect of light and how it creates tone.

To be able to distinguish different types of tone and use this knowledge to make a tonal study of the face.

Description : Students made a tonal drawing of their face using torches and mirrors. Firstly, students had to fill in a brief worksheet investigating tone, to see could they distinguish the various tones before they actually began to make a tonal study of their face.

NOGEN MOGEN

Obtentines of Control on Stadigits Magnilegge of the programmed of the programmed of the control of the programmed of

Steam Thread

Specifies of relation an understation of the credit of 1916 and 1998 at a second sec

Week Four :

Objectives : To learn the procedures involved in completing a worksheet based on the Andy Warhol exhibition in the Irish Museum Of Modern Art. **Description** : The group were taken to IMMA where the curator went through the Andy Warhol exhibition. I had chosen some works of art by Warhol which I wanted the curator to discuss in detail as they were related to the worksheets. The visit to IMMA was specifically to be used as a support study and students completed a worksheet. Students had to document their visit to the museum in their support studies copy. Therefore, as a result of this visit, the class became familiar with the artist and his works.

Week Five :

Objectives : To recap on how to distinguish the various types of tone learnt in last weeks class.

To break down the face into simple shape.

Description : Class began with a discussion on the visit to IMMA. Then students made a tracing of the tonal study of their face breaking it down into shapes by defining all tonal areas. Students then had to complete a worksheet investigating various colour studies of the face, trying out a variety of colour combinations.

Need Meedia

(1) (2) in the force set the encession in the set of the set of

TENT REEN

Objectives To receive on how endisting the his work designees of tone heard in terd buecks dates to break now the fider and Sarak analysis Or activities and the fider and Sarak analysis Or activities and the fider and Sarak and the view to MMA Then the ments more there on the facts and the the last to MMA Then the ments more there on the facts and the the last to MMA. Then the ments more there on the facts and the the last to MMA Then the ments more there of the last and the the last to the ments of the the second form and second the last and the last to the last the ments and the total second to the last of the facts of the term terms of the last of the last of the facts of the fact of the last of the last of the last of the last of the fact of the terms of the last of the last of the last of the fact of the fact of the last of the last of the last of the fact of the fact of the terms of the last of the last of the last of the fact of the fact of the terms of the last of the last of the last of the fact of the fact of the terms of the last of the last of the last of the fact of the last of the terms of the last of the last of the last of the fact of the last of the last of the last of the last of the terms of the last of

Week Six :

Objectives : To develop an understanding and appreciation for colour theory, through a recaping exercise and colour exercises.

Description : In this class students had to refer back to last weeks colour studies, chose one they like and paint a large A3 size study of their face broken down into simple shapes. Students had to refer to worksheets as a guide.

Week Seven :

Objectives : To introduce students to the whole area of screenprinting. Students become aware of the materials and techniques involved in tracing and cutting out their first colour stencil.

Description : To begin class I introduced students to screenprinting and explained the procedures involved. Students then stuck up their large A3 size colour study of their face and traced out all areas that were one colour i.e. all yellow areas. Students then cut out these areas using a blade - this was their first stencil.

Mais As M

Description : united datas etilicante nod (a mise medicanted : non-to-sed activity of the sub-sector and participation of the state state activity of the sub-sector sector and participation of the sector activity and the sector of the sector of the sub-sector - Sub-sector activity to the total to solve the sector activity as a

TONGE SHOP

Objectives 1 clairoduce subletics torizes now a tea or ecreption the knowns recome events of the main service and technices growings and technic div. curding out their metalogical services and technical services and the Decemption of the robust services and the service to contract the server of the services of the service of the service of the server of the server of the services of the service of the service of the server of the server of the server of the services of the services of the service of the server of the server of the server of the server of the services of the service of the server of th

ACTE CONTRACTOR STATE

Week Eight :

Objectives : As an understanding of screenprinting has been gained, students must now apply this knowledge while printing and cutting out the remaining stencils.

To improve students vocabulary relating to the screenprinting process.

Description : Students continue with the screenprinting process. They trace out the second colour and cut this out using a blade. This is their second stencil. While students were working others were taking turns printing their first stencil. By the end of this class everyone had their first stencil printed and were ready to print their second stencil in next weeks class. All students had to print on lengths of wallpaper and they had the background colours already blocked in.

Week Nine / Ten :

Objectives : To reinforce students knowledge of the screenprinting process that they have explored in previous lessons.

To reinforce terminology etc. learned throughout the scheme.

Description: Within these two classes students traced out and cut out their third stencil and again took turns to print their stencils. The project was now complete.

Mees Elimit

Objectives we at understance of screen million labor son golded automa cast new gody mails cowfedge while printing and othing out the restantion stendto

Comprove blights consolier intering it in some property interesting provess first description which its continue with the society ming process first mate out the score exponent data this out using to blace. The first scored et into score exponent working of one were taking futus scored et into score on out of the used working of one were taking futus causing the first sector of the unit of this class everyone had their first score of the first sector of the unit of this class everyone had their first of each of students and the one of the their score were taking first and the first sector of the one of the their score were and the first of each of students and the one of langues of the base everyone had the first of the students and the one of langues of the score were and the first he dates of the students and the one of langues of the score of the first he dates of the first students and the first score of the first he dates of the first students and the score of the score of the first he dates and the students are the score of the score of the first he dates are the students and the score of the score of the first he dates are the students and the score of the score of the first he dates are the students and the score of the score of the first he dates are the students and the score of the score of the first he score of the first he dates are the students and the score of the score of the first he score of the first her score of the first her dates are the students and the score of the score of the score of the first her score of the score of the first her score of the score of the score of the first her score of the first her score of the score of the

LONG REPORT NOT A

Derectives in a contract our denial has readed as an energy of the rest of the

The following are some examples of support studies I used throughout the scheme;



ILLUSTRATION 1: Line drawing of the face by Picasso.









ILLUSTRATION 3: Slides used as a form of support studies.





ILLUSTRATION 4: Self portrait by Andy Warhol.





ILLUSTRATION 5: Silk screen print of Marilyn Monroe by Warhol.




ILLUSTRATION 6: A portrait by Peter Howson.





ILLUSTRATION 7: Warhol has used various colour combinations to convey the image of the face.



Background information on IMMA.

The Irish Museum Of Modern Art has much to offer in terms of exhibitions, workshops, education programmes and so on. The natural and historical context of the Royal Hospital Kilmainham and its grounds are a rich resource and attract many artists to work at IMMA. In 1991 IMMA set up a Education and Community Programme which has been developed with the intention of creating and increasing access to the visual arts and the programme is geared towards older people, children, young people, womens groups, schools, general visitors, students and scholars. The programme also explores aspects within the formal education sector and with community development structures both locally in the Dublin area and part of a regional programme. This programme has continued to operate on many levels and already there has been an increase in the second levels schools sector and the programme will continue to respond to specific needs of these young people and their teachers to insure that they both can benefit from visits to the museum.. One of IMMA's major strengths is its education and community programmes aimed at providing the public with opportunities for direct contact with artists encompassing a wide range of materials and ideas. IMMA also offers an artist - in - residence programme which allows people to witness how other artists work which can be very exciting for students to see. IMMA attempts to reach and serve those sections of the general public who do not normally choose to come to museums and it is up to the teacher, like myself, to make use of these

AMM no pollarmothi bhumpion on MMA.

services offered by a museum establishment and IMMA is a perfect example.

Andy Warhol Exhibition in IMMA.

IMMA approaches this sector of the school population by providing lectures and workshops on current exhibitions. In November 1997 a exhibition by Andy Warhol "After the party, Works 1956 - 1986", was held at the museum. Sponsored by ACC Bank, the exhibition was the first major showing in Ireland of Warhols work. It comprises some hundred works, drawn mainly from the collection of the Andy Warhol Museum in Pittsburgh and includes early drawings from the 1950's as well as better known iconic works from the 1960's and 70's such as the Marilyn, Jackie, Mao and Cambells Soup can paintings. Examples of Warhols Cow Wallpaper, Cloud Pillows, Disaster paintings and a range of source material are also included plus a series of angel and cat drawings by Warhols mother, Julia Warhol.

This exhibition provided an opportunity for second level students to study a modern American artist who was among the innovators of Pop Art. In addition, IMMA provided a detailed catalogue to provide people with information on the artist and his works which were being exhibited. In contained information and illustrations of his work on exhibition in the museum, exploring different stages of the artists development and outlined aspects of his artistic and aesthic education. Therefore, I am going to discuss the importance of using IMMA as a support study by implementing an actual museum / gallery visit to the Andy Warhol exhibition natural energy for the transformation of a section of the sector in the sector of the

And Warhol Exhibition in IMMA

And a cost of a second field of the second development of the second biology of and a cost of a context of a biology of the second of the second biology of another the second of a second of the s

Contrar store and y a new managements in a version of the second strain of the second stra

presently be shown. This will relate directly to one of my teaching practice schemes to reinforce the idea of using IMMA as a support study within the classroom environment and will be interwoven within all work relating to the project . One cannot emphasise enough the importance of support studies as it strengthens the students learning ability and a museum or gallery visit can be an excellent form of support study. presentity be shown if the will treate directly to one of him to realing precision of the science in administration of the science o

daliane vielt can he an excellent form of support offudy.

CHAPTER FOUR

Planning a visit to IMMA.

Teachers organise school visits to museums and galleries to stimulate, develop skills and observe original works of art. The outcome of a museum visit will depend on how well it is planned. When preparing for a visit one must not leave everything till the last minute as there is so much to arrange and organise and if things go wrong one needs time to put them right again. The type of visit you will plan will depend a lot on what you are studying, the age of your students, the reasons for making the visit in the first place and so on. No visit is simple and it is not easy to organise. Much of the work is involved in all stages, from start to finish. A visit to the museum is an opportunity to see new places outside the school as well as to develop communication skills. Pupils find the museum / gallery exciting and visually stimulating and with preparation can gain a great deal from a visit which acts as a crucial stimulus to school work. As Graeme Talboy states ; "Good planning, preparation and plenty of expierence is the key to success". (1)

While planning a visit to a museum / gallery most people follow the usual procedures of ;

(1). Preliminary preparation for both student and teacher

(2). Actual visit to museum / gallery

(3). Follow up work

CHAPTER FOUR

Adduit of their spanning?

1. Explore the second character of gets where the second secon

vinto Lanoro a vetroa e nesedativ dodervinosi popolo Slovning Uslav procidires o

- service of the service for and started and leading
 - ryellar unnaburn distriktura (SI
 - Store of white First

This structure has been recommened by both Woodward and Talboy and is usally used by most organised and well prepared teachers. For the remainder of this chapter I will discuss the procedures involved in organising a visit to IMMA using the above guidelines.

Preliminary work for a museum / gallery visit.

Prior to the visit with students, I made my own preliminary visit to IMMA were I could see the exhibition for myself. This visit was of great importance to me as I was able to become familiar with the museum and its facilities. It also gave me a chance to personally introduce myself to the museum's education officer, Ann Davoren whom I found to be very helpful, supportive and approachable.

This preliminary visit allowed me to finalise all the arrangements for the visit. I was able to confirm the visit and book the curator, find out about where we would meet the curator, where they would leave their coats and bags, where toilets were and learn the basic layout of museum. This preliminary visit also gave me time to look at the exhibition by Andy Warhol and gather information in order to design my own worksheet for this particular class. As a result of this introductory visit I was a lot more focused, happier and content about bringing a class out on a trip to IMMA. As G. Talboy states ;

"The success or otherwise of a visit depends to a large extent on the planning and preparation. A well planned visit for which (i) his structure rues ages recommended by both Wordward and Labovard is usable rectiliby most organised and well prepared leaders. For the entended of this, chapter is will discuss the procedures (protocol) in entended of this, chapter is will discuss the procedures (protocol) in entended of this chapter is will discuss the procedures (protocol) in

celeminary work for a nuseum ligitliary yields of

Prior to the version with students, a made any monoral minanwell to MMA were reculd searing an exhibition for myrah. This was was of meaning or proce to the as these able to become contribution the hosseum and us rachings to not gave the properties to nervorally includure investition also materines achieves that offices. And Device normal frame to be very to materines achieves. And Device way in the to be very

amangamenta (or the Visio 1 was able to confirm the visit a or occis the constor firm out about where we would meet the curator where they would heave instructions and begs, where tables were content the bescheyour of mussume The meanment visit also gave meal then to tools at the excitation by Ann. Washor and gathes information in order to dools at the excitation worksheet for this period blass. As a result of the initiation with what and more than the order that blass. As a result of the initiation with what to time A as over the contract acoust end of the source of a the to time A as 3. Taboy states in the stand browning a datas out on a the to time A as 3. Taboy states in the stand browning a datas out on a the to time A as 3. Taboy states in the stand browning a datas out on a the to time A as 3. Taboy states in the stand browning a datas out on a the terms of the states and the states and the states out on a the terms of the states in the states as a state of the states out on the states as the states out on the states out on the states and the states out on the states out on the states as the states of the states out on the sta

Caldwine period have a monthlecold prediction of the provider of

everyone is well prepared is far more likely to be a success than one that is ill concieived and hastily organised." (2)

Preliminary work within the classroom .

Pupils also need to be prepared for the visit. You cannot visit a museum / gallery with the belief that pupils will understand merely as a result of the visit. Pupils will need to know why they are going to the museum / gallery and what work they have to get done during the visit. This means that adequate classroom preparation is of great importance. If their is inadequate preparation then the museum / gallery trip can be of great disappointment for both pupil and teacher and the trip may not be enjoyable with no learning achieved.

The following project was under taken in Mount Temple Comprehensive School, Malihide Road, Co. Dublin. This second year group were mixed and of mixed ability. There was a total of twenty four students and the scheme ran for the duration of eight weeks and incorporated a visit to IMMA. As a preparation for the visit, the class was introduced to the artist Andy Warhol by means of a slide show and I also gave them information on the exhibition hoping to build up their curiosity and anticipation. It was up to me the teacher to give them direction to help them focus on what they were going to be asked and to clarify the tasks that I want them to complete during visit, like the worksheets. This class was very important as I introduced them to the overall project so they could see how the museum trip would relate to their work. everyona is kell propared is tay more wery to us a succes

(12.) " perinapio vidanti bus per and brashi ang masa " (12.)

Preliminary work within the classroom

Prois can sed to cert and the part of the visit from the set of the sentence of the set of the best of the best of the set of the se

Concretences School: Mailholo, Read, Co. Esciin. This second resprouptive finance and of rinord adino. There was a tota of fivonty foshapents and the accornal toth for Tos-duction of leight weeks and neorholdid a view of MMA. As a presention for the visit, tos class via introduced to the setar new Wartol or means of a skile snow-and 1 ale quee from team and or the evolution homo, to order and too according to the rest of the tracted or the date snow-and 1 ale and tother to the action the evolution homo, to order to the rest of the first and the tracted of the snow-and 1 ale and tother to the setar of the tracted or the snow-and 1 ale and tother to the setar of the tracted of the snow-and 1 ale and tother to the setar of the tracted or the snow-and 1 ale and tother to the setar of the tracted of the snow-and 1 ale and tother to the set of the tracted or the the snow-and 1 ale and tother to the set of the tracted of the snow-and 1 ale and tother to the set of the tracted or the tracted of the snow-and the tracted of the start of the snow of the tracted of the ale of the set of the start of the snow of the snow of the tracted of the start of the snow of the snow

Mowhen or the end of the lithe shares where a price

The museum / gallery and the work carried out there will be the bases for the project back in the classroom which is based on self portraiture. The students will have to investigate the simple proportions of the face and draw their own self portrait to create their own personalised wallpaper using the screenprinting method. They must also keep in mind the work of Andy Warhol and look closly how he used images of his own face and faces of famous icons like Marliyn Monroe or Jackie Kennedy. Students will collect information through worksheets completed in IMMA and will then be introduced to screenprinting by going through various stages - design process.

Actual visit to IMMA.

On arrival at IMMA the students gathered in the audio visual room and watched a short ten minute video on the history of Royal Hospital Kilmainham while waiting on the curator to arrive. The following is a short account on the visit to museum ;

After the video was over I went over all the rules and regulations that had to be kept like on messing, no touching, no running, no food and so on .Within this room I distributed the worksheets which I designed and read through them to make sure no one had any difficulties.

a jaw-avad jug halast show entrons har or there will have

In Lutes for the prosect bac, acting deservant what is thego on set complete and the succerts will have to investig to the annual victorians of the term and the succerts will have to investig to the must victorians and residence using the recentantical method. They must also keep in indidia work of Anu, Waltor and took cleak in the residences of his two the set and faces of femore from the cleak in the residences of his two closes and faces of femore from the cleak in the residences of his two closes in the other increasing and the residences of his two closes in the other increasing and the residences of his two closes is the other increasing and the residences of his two closes is the other increasing an another is the residences of his two and wall then the inforduce to accessing the residences of his two closes is here and the residences of accessing the residences of his two closes is here and the residences of accessing the residences of his two closes is here a pressing the residence of the residences of the residence and wall the tother increasing the residences of the residence of the residences of the residences of the residence of the residences of

Actual visit to MMA

The street and beauty in the marker of more setting with the product of

The following are photographic evidence of the students at work;



ILLUSTRATION 8:.Students at work in the museum.





ILLUSTRATION 9: Students at work in the museum.



This was very important as the worksheets were the main means of bringing back information into the classroom situation. We then meet the curator who brought students on a guided tour around the Andy Warhol exhibition focusing only on a few pieces relating to the overall project images of the face. Some students were answering their worksheets as they went along , however, I would not give out any colouring pencils till after the talk with curator was over as it would only break their concentration.

The curator was excellent. She spoke very clear, precise and used simple vocabulary for this second year group and she was very good at relating to students which made the talk very successful. After the guided tour students asked questions and were very interested and responsive. They were then given thirty minutes to wander around the "After the Party " exhibition and fill in worksheets by referring directly to the work on show at this exhibition. At this particular time I distributed packets of colouring pencils to students in order to complete worksheets in more detail. I constantly walked around the exhibition making sure students were completing worksheets and not messing. I was able to take photos of students at work which was a great way of recording the day. I was a little nervous about students getting worksheets complete but most of them completed the tasks outlined by the end of the thirty minutes. Students were very well behaved during their one hour fifteen minute visit to the Irish Museum Of Modern Art. The class meet in the main reception area at eleven thirty and we made our way back to bus and arrived back in Mount

40

An and the second as the workers were as the means of the means of

Temple at a quater past twelve just in time for their next class. From the reaction of students on the bus they seemed to have really enjoyed themselves and their previous attitudes about the work of Andy Warhol had now changed and most students liked and appreciated his art work as a result of the visit.

Follow up work back in the classroom.

The follow up work is what is carried out back in the classroom situation and how the information gathered in the museum is integrated into the scheme of work is very important. This stage is a vital part of the project and it must be put to good use. The museum / gallery work should be an obvious development in the classroom and students should be working from both previous classroom work, worksheets and secondary sources like slides, postcards, visual aids, support studies and so on. To begin the class we talked about the visit and discussed what they enjoyed most - we evaluated the visit to IMMA. Within this discussion we talked about the artist and his work. Students were very responsive and their opinions and toughts varied. With regard to the worksheets the majority of them were fully completed and we used these to refer back to while discussing the artist and his work. (Appendix 2)

It is very important to consider this visit as a preparation for the follow up work. It gives students the opportunity to express their feelings about the visit, the artist and his work. The success of the visit was emphasised back at Mount Temple, where the excitement and enthusiasm A source of a gradie deer because qual in more to alter a next open. From the control of the press of the press of the control of the press of the press of the control of the control

collow up werdebaldk in the diassioners.

An althout phuality of the unormaticity phuloted of the measure of the end book the unormaticity phuloted of the measure of the moment phuloted of the measure of the moment in the measure of the measur

wenth each work. It groes audents the proportion to express their work which entry as which each to a second entry provide the work. The all checks of the visit was fitted and provide the original proportion where the excision and and entry all all of the students to finish the project by continuing onto the next stage was overwhelming. The students involved in this visit gained a lot from the expierence. Students now have a good understanding of the Andy Warhol exhibition and have extracted the relevant information needed from the exhibition and are ready to begin relating it to their own personal work. The worksheets will now be used as a form of support studies for the rest of their work to which they can refer back.

Students referred back to their work from previous classes which took place before the museum trip. Within these classes students learnt about the proportions of the face, discussed protraiture and looked at artists who dealt with self portraiture. Students then drew their own self portrait using a mirror and brought this a stage further by making a tonal study of their face using a mirror and a torch to create a variation of tones. The visit to IMMA allowed students to see how Andy Warhol treated the image of the face and students were ready to move onto the next part of the project. This involved tracing over their tonal studies of the face, breaking it down into shapes by defining all tonal areas. Students then had to complete a worksheet investigating various colour studies of the face, trying out a variety of colour combinations using a maximum of three colours only. Students then had to select their favourite colour study and paint a large A3 size study of their face broken down into simple shape and had to constantly refer to worksheet. Students were getting use to the idea of worksheets as this was their third one so far.

The next part of the project involved students being introduced to the whole process of screenprinting which was completlely new to them. At this point I introduced students to screenprint artists like Bridgit Reilly, Beryl Cook and Moira Wills. This was very important as I wanted students to be introduced to a large selection of artists within this project and wanted to keep reinforcing the idea of support studies and its importance. Students then stuck up their A3 size colour study of their face and traced out all areas that were one colour, for example traced out all areas that were yellow. They then cut out these areas using blades and this was their first stencil. I was a little nervous that students would cut themselves or mess with the blades but they were very well mannered and no messing took place. Students continued to cut out each colour to make their stencils and printed till the project was complete, going through the whole process of screenprinting and I was constantly introducing new vocabulary based on the overall process. Students were very disciplined as their is a lot of preparation and cleaning up involved in this process from tracing to cutting out, from printing to cleaning the screens. Students were very organised and all students made an effort when it came to tidying up. Students achieved a high standard in their practical work and I was very happy with their overall performance and behaviour.

ed emetude bevic. I below and to nea tran white



ILLUSTRATION 10:Student making a tonal drawing of their face.





ILLUSTRATION 11: Student painting the large A3 size portrait of their face.








ILLUSTRATION 13:Student tracing out stencil.





ILLUSTRATION 14:Students cutting out stencils.





ILLUSTRATION 15:Students at work on their screenprint - dragging the ink.





ILLUSTRATION 16:Students at work on their screenprint - lifting the screen.

Č



ILLUSTRATION 17: Examples of students first stencil prints.



52 ILLUSTRATION 18:A finished piece of work.

FOOTNOTES

(1)Graeme Talboy,<u>Using Museums as Educational Resources</u> (London:Ashgate Publishing Ltd.,1996) p.47.

(2)Ibid.,p48.

FOUTDOF

A Gradmic TalizorQuicand Musequity at EditLatitud, Resources -(condem Ashricta Publichery Etal. 1995) p. 47.

(2.4b/d. n.49

CHAPTER FIVE

Evaluating museum scheme.

Now that the museum scheme is complete I can look back at the overall project and feel satisfied with both my own performance and the students performance. The scheme was designed for the duration of ten weeks and it was fully complete within this period of time. Likewise, each lesson I devised set out specfic objectives which I wanted to achieve and successfully did so. Throughout the project students were well behaved and put in a lot of effort and completed all the tasks that were set out. By students giving this commitment and full participation within class time it allows me to fullfill the objectives within each lesson and the overall aim of the scheme is then achieved. I was happy with my own performance throughout the scheme as a lot of organising and preparation was put into this project from arranging a trip to the IMMA, to collecting mirrors for students to work with to designing worksheets.

From the beginning to the end of this scheme I incorporated support studies in all lessons. Not only was this second year group introduced to a museum visit as a form of support studies but also learned about keeping a support studies copy relating to the overall project. In this copy students had to document all the processes involved in this project and had to gather relevant information relating to the project. This was the first time these students worked with support study copies and most of

AVIA NELANO

superiors and second builts appress

them really enjoyed working on them. They stuck in leaflets from IMMA, glued in photos of themselves, drew in copies, stuck down magazine clippings and anything else they found would relate to project. All copies were very original and unique and students got very involved in keeping copies up to date. Students enjoyed working with this new idea and began to understand the importance of support studies and how relevant they are within the Junior Certificate Art, Craft and Design Course.

Not only did students maintain a support studies copy but I constantly introduced new support studies within every lesson I taught. Overall the importance of support studies and its uses played a large role in my teaching practice and this is highlighted throughout my teaching practice notebooks, as I wanted students to be introduced to a large broad range of artists, designers and art movements. And from reading my disseration so far one can see how I am trying to illustrate that "support studies" are a vital part of any project and should be continous throughout and should be "an integral part of the learning expierence". (1)

Conclusion.

Now that I have come to the end of my disseration I hope you the reader can see the importance of support studies as it is a vital part of the Art, Craft and Design course and this is explored in detail in chapter one. There is not enough emphaise on support studies and it is up to the teacher to keep introducing students to the work of artists either past or present which relate to the topic they are studying, then support studies will them that a cover work of an area of the values and an earlest nom twick gried in metres of the relatives, drev, in cookes, steph drive, dragenape supprints and a value or plactates mend would make to prinest. All econes were not or ease statems and another working with this neurower and the string to ease statems and end working with this neurower and the statementation or exclusion or exception within the methods and the statement of the importance or exception to be and the methods and the statement of the importance or exception to be and the methods and when the to be a statement or exception to be and the statement of the statement of the importance or exception to be and the statement of the statement of the importance of the to be and the statement of the statement of the importance of the statement of the statement of the statement of the importance of the statement of the statement of the statement of the importance of the statement of the statement of the statement of the importance of the statement of the statement of the statement of the importance of the statement of the statement

objectantly introduced now early of and as which much tessors 1 targing Overelining monored of suggert and estates of the tess payers field a tradition we tesoring precise and this is highlighted moushed on teaching are tesoring to thanked suggerts to be introduced to a large tradit of an enters reserved; and all movements for a introduced to a large tradit notione or anists reserved; and all movements for a large tradition mu assession as ten one enclosed suggerts to be introduced to a large tradineedees are antial part of any project and stored and the continues that featibility and any another on integral part of any project and store rules (c).

Conclusion

In react cag seaths intro tates of subtract studies as the a vibili bad of the react cag seaths intro tates of subtract studies as the a vibili bad of one and there is not should be anticide to subport studies and this to the one inera is not should be anticide to support studies and this to the ineration base mudicing students to the work of arrest externotes of one ineration base mudicing students to the work of arrest externotes with one ineration base mudicing students to the work of arrest externotes with one ineration base mudicing students to the work of arrest externotes with one in a support of a students of a students of the students of a support of and one in the students of a students of the students of a support of a student of the student of a student of the student of a give them a greater insight and understanding about this specfic subject. The areas of support studies should be explained to the students at the beginning of a project, that is research work, preparatory drawings and also relvant art historical references and influences. It is important to see support studies as being an essential part of the scheme. It should not just become homework but should be given a structure within the scheme. A folder, copy, notebook, scrapbook, etc should be designated to the area of support studies so that students have a evident place in which to put their research work, drawings, expierments, pictures, information etc. This should be checked regularly by the teacher to see if students are gathering work.

Throughout my dissertation I was concerned with the integration of support studies into practical Junior Certificate Projects and was referring specifically to utilizing galleries and museums. I wanted to emphaise the importance of bringing students on a trip to one of these establishments as it is a primary source and a good form of support study. These places of interest contain marvellous pieces of work and have so much to offer in terms of services and facilities for both teacher and students, and I have discussed this in chapter two. The main reason for visiting a museum / gallery is for pupils to see for themselves original works of Art, Craft and Design. Visits must be made for a real purpose and the expierence of art should be broadened and by seeing an artists finished work through visiting galleries, musuems, workshops, etc is one way of doing this. Specific areas which were developed were vocabulary of words

A sectored the ensite of support studies, should be explained to the education of the ensite of support studies, should be explained to the education of the ensite of support studies, should be explained to the education of the ensite of the education of the ed

Through a function of the second of the concerned with the promin-mention of a pro-ferring to the process through the difference in the proand referring aperferring to the balance of other for and difference in the terror of the foranomale with a concerned of other providents and a monitor of the forenergy second of the protection and the terror of the forthread places of interest contain manualious concernes of work and barle, so thread places of interest contain manualious concernes of work and barle, so another is and the contain manualious concernes of work and barle, so another is and the terror of secondes and facilities for collinated and another is and the decrease of the interest two is the quart terror work and the decrease of the interest two is the quart proended of the decrease of the interest two is the quart prosecond surgery of the decrease of the interest two is the quart prosecond surgery of the decrease of the interest two is the quart prosecond surgery of the decrease of the interest two is the quart prosecond surgery of the decrease of the interest two is the quart prosecond for any decrease the prosecond is decreased by a prosecond is the terror of the prosecond is an another of the second minimate of the terror is appeare decrease and the second minimates interedtion of the decrease with the second minimates and the terror is the terror of the terror is appeare decrease with the second minimates and the terror of terrors with terror appeared and the terror of terrors and the terrors intered terror is appeare decrease with the appeared minimates and the terrors in the terrors and terrors appeared to the terrors in terrors appeared to the terrors appeared to the terrors in terrors appeared to terrors in terrors appeared and terms which were relvant to the scheme. Students had developed a vocabulary in order to discuss their own work and the work of others. They also developed the confidence, belief and ability to speak more openly about their work and discuss other students work in a critical manner. The overall scheme of work and support studies were used as a means of obtaining vocabulary and critical skills which is an important part of the Junior Certificate Syllabus. This has been discussed in both chapters three and four.

In conclusion, I personally found the museum scheme achieved a large amount of results and proved that support studies are an invaluable and precious part of the Junior Certificate Syllabus. Support studies opens up several opportunities for both teacher and student. Teachers are recognising the importance of taking their students out of the classroom and into other learning environments. A visit to the museum and in this case a visit to The Irish Museum Of Modern Art was an opportunity to see new places outside the school as well as to develop communication skills and act as a major stimulus to school work. One cannot emphaise enough the importance of support studies as it strengthens the students learning ability and a museum or gallery visit can be an excellent form of support study.

active de l'alge articul in ravius e la covera d'al a support studies aux sa active de l'alge articul in ravius e la covera d'al support studies aux sa muches e etch impous con el die dunier. Emilieste Sylebus, Support d'unes ducis up several opportuncies ter com tractie and student, commons secure qui several opportuncies ter com tractier and student, commons secure de la muchande of take or tractier tractier and student, al dis case a visi fai fire tractice equitorinania. A visit to the unservent and dis case tractice the tractice of as well us to de selection and the case tractice of a fire tractice of as well us to de selection and and the rade of a sufficient set of activity of a fire selection of the set of the tractice of as well us to de selection entration and the rade of a sufficient set opport strates as distance of the selection of a strates of a strates of a condition of the selection and the rade of a sufficient of a strates as a set of a condition of the selection of a strates of a strates of the selection of a strates of the selection of a strates of a strates of a strates of the selection of the selection of a strates of a strates of the selection of the set of the selection of a strates of a stratege of the selection of the set of the selection of the set of a strates of the set of the selection of the selection of the set of a strates of the set of the selection of the set of the selection of the set of the se

Volte Bonou

FOOTNOTES

(1)Department of Education,<u>National Council for Curriculum and</u> <u>Assessment,"Junior Certificate Art, Craft and Design</u> <u>Guidelines for teachers"</u> (Dublin:The Stationary Office,1992) p.3.

POOTNOTES

Department a Education Actional Council for Europation and Assessment Junity Confifteate, Art Craft and Depart Ormatures for factous (Dublin The Stabonary Office AP34)

APPENDIX ONE







and international fame. Warhol is identified with Pop art, sometimes being credited with its invention. Pop art referred to art which took its imagery and techniques from the world of advertising and mass of media. It rejected distinctions between "high" and "low" art, that is between Fine art and commercial art.

During his working career, Warhol used a considerable number of different methods and media and, in addition, created the same subject in different media and by different methods. He produced memorable single or multiple images of popular icons, such as Marilyn Monroe, Liz Taylor and Jackie kennedy and of familiar everyday objects such as his famous Campbell soup can series. In general, Warhols paintings were usally block silkscreen on hand-coloured backgrounds His prints though, were normally coloured screen prints.

Over the course of a 30 year-long career, Andy Warhol transformed contemporary art. The power of his work comes from its concentration on fundamental human themesthe beauty and glamour of youth and fame, material culture and the passing of time and the presence of death.

Warhol died in 1987 while indergoing a minor surgical operation. His funeral, like much of his life, was a major media event. He is considered one of the most important artists of the late 20 th century.



7 NAME : SCHOOL : CLASS : DATE : JACKIE KENNEDY (1964) - Jackie Kennedy dominated the media after the assassination of John F. Kennedy in 1963. Warhol became fascinated by her and by the media obession with her. His images were taken from a life magazine special Feature on Jackie after the death of her husband, when she became a symbol and Focus of the Americas Grief. Silkscreen enabled Warhol to repeat the images again and again which murrored the constant repetition of them . 1 Just one image From this series of eight. in the media. • What medium was used to make these prints ? · List the colours used here. · Each print is made up From two colours. In your opinion, which colour combination works best 4 why? · Are all the images the same? YES NO · Are the quality of the images clear ? YES (tick appropriate box V)



• Describe the various expressions of Jackie Kennedy's face.

MARILYN MONROE (1960'S)

Warhol became interested in Monroe as a subject after her suicide in Aug. 1862 when the media was full of images of this glamourous and tragic film star. In his silhscreened images of her, her features appear familiar and sexy but distant. and removed the blank background helps to suggest the idea of Monroe as an "icon" separated from reality.



· Describe the expression/mood on Marilyns Face.

 Andy Warhol has taken the image of Marilyn as an "icon" of her time. What other famous icons has he dealt with in this exhibition?

The image of Marilyn Monroe has been used in another work by Warhol. Find this piece and compare both pieces with each other. (for example, expression, colour, medium, shape ctc.)

COW WALLPAPER (1966) . This room gives a light hearted atmosphere to the (A) gullery which is in complete contrast to the seriousness of his other works. The cow has been depicted in art since the 17th century but usally in pastoral landscape. The image of the cow looks very isolated (B) against a bright yellow background. The image of the cow was found by his assistants and the wallpaper was made by manufacturers. • As you walk into the room with the cow wallpaper, what is your reaction. • What type of pattern has Andy Warhol used here? The "motif" of the single head of a cow which makes up the wallpaper design looks very isolated. How does Warhol make the Chairman Mao wallpaper look more Free Flowing and not as isolated. Fill in box A, B. and C. by referring to original wallpaper For guidance. Then add colour to the three rectangles referring to original colours. (Must be accurate by and neat.)



CHAIRMAN MAO (1972)

This large scale portrait of Chairman Mao was exhibited against special Mao Tse Tong Wallpaper. He chose Mao because of the current interest in China which President Nixon visited in 1972. Mao's image was shown all over China and it was a sign of Communism in the West. While the features of Mao have been silkscreened onto Canvas, Warhol has painted over the image in parts using a variety of expressionist brushstrokes.



Describe what you see the minute you walk into this room.

· Would you use this type of wallpaper in your home, give a reason for or against it ? ____

Comment on Andy Warhols : choice of colours.

 How has he modified the portrait into a "motif" suitable For wallpaper?


SELF PORTRAIT (1986) Warhol returns to his favourite. theme - himself as " celebrity". He belonged to an era in which the artist became a celebrity which he loved. He lived a very Flamboyant lifestyle and was constantly socialising. No clues to the artists private self are contained in his portrait. A silk screened photograph of Workol wearing one of his numerous wigs which he used throughout most of his adult life. • What medium was used ? • Describe Warhols Facial expression. • What Feature is most dominant in bus portrait? • What colours were used in this portrait? · Colour in the outline of Andy Warhols portrait (above) by referring to the colours found on original piece. · Did you enloy the exhibition of Andy Warhols work and give reasons for your answer.



Out of the whole exhibition chose your favourite piece and make a drawing of it. (using colour) in the space below. • Describe the piece in your own words. Why did you chose this piece?



APPENDIX TWO

NAME: Pamela Rigan school: Mt Temple Malahide CLASS : 2B Rd DATE : 22.1.98 Dublin & JACKIE KENNEDY (1964) Jackie Kennedy dominated the media after the assassination of John F. Kennedy in 1963. Warhol became fascinated by her and by the media obession with her. His Images were taken from a life magazine special Feature on Jackie after the death of her husband, when she became a symbol and Focus of the Americas Grief. Silkscreen enabled Warhol to repeat the images again and again which mirrored the constant repetition of them 1 Just one Image From this series of eight. in the media. · What medium was used to make these prints ? MotogRaphic Screen Printing · List the colours used here. Dark Blue White 10llow Light Rule Gold _ GRULY · Each print is made up From two colours. In your opinion, which colour combination works best 4 why? Gold + Black you can see more detail and it looks Beller NO · Are all the inages the same? YES · Are the quality of the images clear ? YES NO (tick appropriate box /) (1)



Describe the various expressions of Jackie Kennedy's face. sad, Happy, Depresped, Confuned MARILYN MONROE (1960'S) Warhol became interested in Monroe as a subject after her suicide in Aug. 1862 when the media was full of images of this glamourous and tragic film star. In his subscreened images of her, her features appear familiar and sexy but distant. and removed the blank background helps to suggest the idea of Monroe as an "icon" separated from reality. Describe the expression / mood on Marilyns face she's happy and then to look sedu Andy Warhol has taken the image of Marilyn as an "icon" of her time. What other famous icons has he dealt with in this exhibition? Liz Taylor. The image of Marilyn Monroe has been used in another work by Warhol. Find this piece and compare both pieces with each other . (for example , expression , colour , medium , shape etc) The Gold is cleaner. She looks sick in the second one the expronon a deficient the as soil but TRIED to LOOK Happy

()



CHAIRMAN MAD (1972)

This large scale portrait of Chairmon Mao was exhibited against special Mao Tse Tong Wallpoper. He chose Mao because of the current interest in China which President Nixon visited in 1972. Mao's image was shown all over China and it was a sign of Communism in the West. While the Features of Mao have been silkscreened onto Canvas, Warhol has painted over the image in ports using a variety of expressionist brushstrokes.



Moo 1972-73

 Describe what you see the minute you walk into this room. Lats of Pulpie circles with faces
In them, And a large portrait of a man

• Would you use bus type of wallpaper in your home, give a reason for or against it? <u>Nor norme</u> <u>uxents a picture of a man in fleur Person</u>

Comment on Andy Warhols: choice of colours. PLS ColourFull and Real tic looking the colour are for very road on the firsthait But the Wall paper is Plane

How has he modified the portrait into a "motif" suitable for wallpaper? Its pright and colorupfiel and a happy effect

(1.)



62 COW WALLPAPER (1966) This room gives a light (A) hearted atmosphere to the gallery which is in complete contrast to the seriousness of his other works. The cow (c)has been depicted in art since the 17th century but usally in pastoral landscape. The image of the cow looks very isolated (B) against a bright yellow background. The image of the cow was Found by his assistants and the wallpaper was node by manufacturers. • As you walk into the room with the cow wallpaper, what is your reaction. you feel happy Because of the Bright colder • What type of pattern has Andy Warhol used here? 2 DRap • The "motif" of the single head of a cow which makes up the wallpaper design looks very isolated. How does Warhol make the Chairman Mao wallpuper Free Flowing and not as isolated. look more 91- Looks Red Spacious. And happy • Fill in box A, B. and C. by referring to original wallpaper For guidance. Then add colour to the three rectangles to original colours . (Must be accurate by referring and neat.) 1-1



SELF PORTRAIT (1986)

Warhol returns to his favourise. theme - himself as "celebrity". He belonged to an era in which the artist became a celebrity which he loved. He lived a very flamboyant lifestyle and was constantly socialising. No clues to the artists private self are contained in his portrait. A silk screened photograph of Worhol wearing one of his numerous wigs which he used throughout most of his adult life.



• What medium was used? <u>Screen Printing</u>

- Describe Warhols Facial expression. He nocks Angry and as if his going to kill someone
- What feature is most dominant in bus partrait? It is faciel expression
- •What colours were used in this portrait? <u>Red + Black</u>
- Colour in the outline of Andy Warhols portrait (above) by referring to the colours found on original piece.
- Did you enloy the exhibition of Andy Workols work and give reasons for your answer. yes. It went theorigh different Stages of Lit



Out of the whole exhibition chese favourite piece and write a drawing of it (using colour) in the space below. • Describe the piece in your own words. It offeels danger and ringer. The clours. are nice and Bright and Shows how the is feeling after Being Shot Why did you chose this piece? Because its like a feeling. (6)



BIBLIOGRAPHY

1. EISNER, ELLIOT ; <u>'EDUCATING ARTISTIC VISION'</u> MacMillan Pub., New York, 1972.

2. HOOPER-GREENHILL, EILEAN ; <u>'MUSEUM AND GALLERY</u> EDUCATION' Leicester Uni.Press,London,1991

3. HOOPER-GREENHILL, EILEAN; <u>'THE EDUCATIONAL ROLE</u> OF THE MUSEUM' Routledge, London, 1994

4. KAELIN, E.F.; <u>'THE AESTHETICS FOR ART EDUCATION'</u> London, Teachers College Press, 1989

5. TALBOYS, GRAEME ; <u>'USING MUSEUMS AS AN</u> EDUCATIONAL RESOURCE' England, Arena, 1996

6. TAYLOR, ROD ; <u>'EDUCATING FOR ART'</u> London, School Curriculum Development Committee, 1986

7. TAYLOR, ROD ; <u>'VISUAL ARTS IN EDUCATION'</u> London, Falmer Press, 1992.

8. ZETTERBERG, HANS, L; <u>'MUSEUMS AND ADULT</u> EDUCATION' London, Evelyn, Adams and Mackey, 1968

OTHER SOURCES

DEPARTMENT OF EDUCATION; <u>NATIONAL COUNCIL FOR</u> <u>CURRICULUM AND ASSESSMENT 'JUNIOR CERT. ART CRAFT</u> <u>AND DESIGN SYLLABUS'</u> Dublin, The Stationary Office, 1992.

DEPARTMENT OF EDUCATION; <u>NATIONAL COUNCIL FOR</u> <u>CURRICULUM AND ASSESSMENT' JUNIOR CERT. ART, CRAFT</u> <u>AND DESIGN GUIDELINES FOR TEACHERS'</u> Dublin, The Stationary Office, 1992.



DEPARTMENT OF EDUCATION; <u>CHARTERING OUR</u> EDUCATION FUTURE 'WHITE PAPER ON EDUCATION' Dublin, The Stationary Office, 1992.

WOODWARDS; <u>'GOLD OF THE PHARAOHS' SCHOOL VISITS TO</u> SCOTTISH MUSEUMS NEWS 3., 1989:3-5

